Author's response to reviews

Title: Impact of Acute Exercise on Brachial Artery Flow-Mediated Dilatation in Young Healthy People

Authors:

In-Chang Hwang (inchang.hwang@gmail.com)
Kyung-Hee Kim (learnbyliving9@gmail.com)
Won-Suk Choi (cram464@gmail.com)
Hyun-Jin Kim (titi8th@gmail.com)
Moon-Sun Im (immoonsun82@gmail.com)
Yong-Jin Kim (kimdamas@snu.ac.kr)
Sang-Hyun Kim (shkimmd@snu.ac.kr)
Myung-A Kim (kma@brm.co.kr)
Dae-Won Sohn (dwsohn@snu.ac.kr)
Joo-Hee Zo (jooheezo@hanmail.net)

Version: 2 Date: 23 September 2012

Author's response to reviews:

September 23th, 2012
Rosa Sicari, MD,
Editor-in-Chief of the Cardiovascular Ultrasound

Dear Rosa Sicari,

I, on behalf of all the authors, would like to thank you and the reviewers of ‘Cardiovascular Ultrasound’ for taking time and effort to review our manuscript. The comments were thoughtful and provided valuable insights into our study. Having reviewed these comments we have extensively revised the manuscript which we now offer for reconsideration. Below are the point-to-point replies to the reviewer comments, with the revised portions were tracked in the manuscript.

We have tried our best to abide by the recommendations and comments of the reviewers. We believe these revisions greatly strengthen our manuscript and hope that the revised version will better meet the requirements of the ‘Cardiovascular Ultrasound’ for publication. We thank you again for your constructive and detailed review, and we hope to hear some good news in the near future.

Sincerely,

Joo-Hee Zo, MD
Division of Cardiology, Department of Internal Medicine
Seoul National University Boramae Medical Center
20 Boramae-ro 5-gil, Dongjak-gu, Seoul 156-707, Korea
Tel: 82-2-870-2214
Fax: 82-2-870-3866
Email: jooheezo@hanmail.net